Author's response to reviews

Title: Relationships between Sexual Violence and Chronic Disease: A Cross-Sectional Study

Authors:

Jeanie Santaularia (jsantaularia@kdheks.gov)
Monica Johnson (mjohnson5@kumc.edu)
Laurie Hart (lhart@kdheks.gov)
Lori Haskett (lhaskell@kdheks.gov)
Ericka Welsh (ewelsh@kdheks.gov)
Babalola Faseru (bfaseru@kumc.edu)

Version: 2
Date: 16 July 2014

Author's response to reviews: see over
July 16, 2014

The Executive Editor,
BMC Public Health

Reference: Submission of an article

Dear Sir/Madam,

Thank you for your initial editorial review of our manuscript. We have included line numbering and page numbers as directed.

Please let me know if there is anything else we need to do in order to expedite peer review.

Thank you,

Yours sincerely,

_______________________________
Babalola Faseru, MD, MPH
Corresponding Author